Author's response to reviews

Title: Serum VEGF levels are related to the presence of Pulmonary Arterial Hypertension in Systemic Sclerosis

Authors:
Andriana I Papaioannou (andriana78@vodafone.net.gr)
Epaminondas Zakynthinos (ezakynth@yahoo.gr)
Konstantinos Kostikas (ktk@otenet.gr)
Theodoros Kiropoulos (ktheod@med.uth.gr)
Angela Koutsokera (ilkoy@otenet.gr)
Athanasiou Ziogas (aziogas@yahoo.gr)
Athanasiou Koutroumpas (thanoskoutr@msn.com)
Lazaros Sakkas (lsakkas@med.uth.gr)
Konstantinos I Gourgoulianis (kgourg@med.uth.gr)
Zoe D Daniil (zdaniil@med.uth.gr)

Version: 6 Date: 27 March 2009

Author's response to reviews:

Dear editor
Thank you for accepting our manuscript (MS: 1730606233212062) Serum VEGF levels are related to the presence of Pulmonary Arterial Hypertension in Systemic Sclerosis Andriana I Papaioannou, Epaminondas Zakynthinos, Konstantinos Kostikas, Theodoros Kiropoulos, Angela Koutsokera, Athanasiou Ziogas, Athanasiou Koutroumpas, Lazaros Sakkas, Konstantinos I Gourgoulianis and Zoe D Daniil) for publication in BMC pulmonary medicine.
As you have requested we have made the following changes:

1. Please remove the authors' qualifications from the Title Page.
The authors' qualifications have been removed from the Title Page

2. Please make the following statement: "This study was supported by the State Scholarships Foundation", which currently appears on your Title Page, in the "Acknowledgements" section of your manuscript.
The statement “This study was supported by the State Scholarships Foundation” was moved in the "Acknowledgements" section of the manuscript.

3. In your abstract, please correct the mistake in the sub-section heading "Results" - there is presently a gap between the 'u' and 'l'.
The mistake has been corrected

4. Please rename the "Materials and Methods" section as "Methods".
The “Material and Methods” section has been renamed as “Methods”
5. Please rename the "Conflicts of interest" section as "Competing Interests". "Conflicts of interest" section has been renamed as "Competing Interests".

Authors have gone through the manuscript formatting checklist and have ensured that the revised manuscript conforms to all of the points.

We are in your disposal for any further clarifications
Yours sincerely
Andriana I Papaioannou, MD